Author’s response to reviews

Title: A Rare But Treatable Cause of Recurrent Chest Pain - Ictal Chest Pain

Authors:

Ching Soong Khoo (chingsoongkhoo@gmail.com)

Dongah Lee (peony1003@gmail.com)

Kang Min Park (smilepkm@inje.ac.kr)

Byung In Lee (BILEE@paik.ac.kr)

Sung Eun Kim (epidoc@inje.ac.kr)

Version: 4 Date: 11 Dec 2019

Author’s response to reviews:

Editor’s comments:1. Changes have been made according to the suggestions. 2. It has been revised. 3. Authors’ contributions have been modified according to the guidelines. 4. Corresponding author. 5. A finalized and clean manuscript has been prepared.